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Data Taking

Day Delivered  Recorded  Efficiency Comment
Lum (pb™)  Lum (pb™) (%)
L1CTT and Muon readout crate cause

7 Nov 296 242 82 downtime. Access to swap CTT board

8 Nov 3.40 2.76 81 L1 Muon PS trip and continued CTT issues.

9 Nov 2.38 2.21 93 Access to replace CTT power supply.

10 Nov 5 95 5 46 23 CTT octant still dc;\;vl?énMuon special runs

11 Nov 305 337 g Master clock problem. Data.logger disk

errors. More muon special runs.

12 Nov 3.52 2.99 85 Access to replace CTT power supply.

13 Nov 2.56 1.98 77 ICD HV trips twice. Silicon bias scans taken.
713 Nov 517 182 84 Record week for recorded and delivered

luminosity.
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DO Significant Events

* 7 November - 82%
— L1CTT problems throughout morning.

* Two sectors disabled.
* Access to replace AFE board.

— Muon readout problem causes downtime at the start of each run.
Thought to be a chip that went bad at the start of the store. Power
cycle fixed the problem.

e 8 November - 81%

— Cal preamp trips and STT readout problems for 10 minutes at start
of store.

— 40 minutes to fix L1 Muon power supply trip.
— L1 CTT continues to have problems.

— One hour L1CAL RunlIb special run taken.
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DO Significant Events 11

e 9 November - 93%

— Access to diagnose and fix CTT.
* Power supply replaced.
* Took four hours and still did not fix the problem.
* 10 November — 83%
— Record DO initial luminosity of 154E30
— Special Muon runs taken.

— Operator error causes 20 min downtime at start of store.

e 11 November — 85%

— Master Clock glitch causes one hour down.
— Failing disk causes datalogger to back up DAQ intermittantly.

— More Muon special runs.
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D Significant Events 111

* 12 November - 85%
— Access to replace power supply for CTT.

* Still unsuccesstul.
— PDT HV trip — could not ramp back up to full.

e Access needed to fix.
e 13 November — 77%

— ICD trips twice

* DO down 30 minutes during owl shift.
e Would not ramp to full after 2™ trip.
— Traced to faulty HV board which was repaired.

— Down for two hours
— Last two hours of the store Silicon bias scans taken.
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CTT Summary

* When the board replacement failed on Monday, possible other issues
were looked at on the Wednesday access.

— It was found that there was significant ripple on the 3.3v power
supply and the supply module was replaced which did not solve the
problem.

* On Saturday a new module with a noise filter was added, but still did
not solve the problem.

* We are looking to try a different type of 3.3v supply, but are awaiting
safety inspection.

— We are also investigating if there 1s a problem with the power
supply circuitry on the board.

* The RunlIb system will use a completely different power supply
arrangement.
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DS Beam Spot Move
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[LICAL Leak found
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e [.ast week we found our

fourth leak 1n our
L1CAL system this
year.

— Between stores this
week the drip was
1solated to the second
lowest radiator.

— The good news 1s that it
was not an old leak, so
our copper putty
solution still works.

Bill Lee



m Daily Data Taking Efficiency | 19 April2002 - 12 November 2005
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m Run Il Integrated Luminosity 19 April 2002 - 12 November 2005
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